Author's response to reviews

Title: Muscle Research and Gene Ontology: New standards for improved data integration.

Authors:

Erika Feltrin (erika@cribi.unipd.it)
Stefano Campanaro (stefano.campanaro@unipd.it)
Alexander D. Diehl (erika@cribi.unipd.it)
Elisabeth Ehler (elisabeth.ehler@kcl.ac.uk)
Georgine Faulkner (faulkner@icgeb.org)
Jennifer Fordham (jennifer.fordham@kcl.ac.uk)
Chiara Gardin (chiara.gardin@cribi.unipd.it)
Midori Harris (midori@ebi.ac.uk)
David Hill (erika@cribi.unipd.it)
Ralph Knoell (rknoell@med.uni-goettingen.de)
Paolo Laveder (laveder@cribi.unipd.it)
Lorenza Mittempergher (lorenza.mittempergher@cribi.unipd.it)
Alessandra Nori (alessandra.nori@unipd.it)
Carlo Reggiani (carlo.reggiani@unipd.it)
Vincenzo Sorrentino (v.sorrentino@unisi.it)
Pompeo Volpe (pompeo.volpe@unipd.it)
Ivano Zara (ivano.zara@cribi.unipd.it)
Giorgio Valle (giorgio.valle@unipd.it)
Jennifer Clark (jdeegan@ebi.ac.uk)

Version: 6 Date: 16 January 2009

Author's response to reviews: see over
To the BioMed Central Editorial Production Team

Tel: +44 (0)20 7323 0323  
e-mail: corepreaccept@biomedcentral.com  
Web: http://www.biomedcentral.com/


Dear BMC Editorial Production Team,

First of all we apologize for our delay.  
Following your previous note, we included formatting modifications, as follows.

We included the names of all authors from the group within the list of authors. We also added a note in the Authors’ contribution defining the members contained within the group and the role of the group within this study.

We checked the references and included all authors names deleting the term ‘et al.’.

We hope that the paper is now ready for publication.

Best regards,
Erika Feltrin

Padua University  
Via Ugo bassi 58/b (Italy)  
Tel: +39 049 827 6165  
e-mail: erika@cribi.unipd.it